Author’s response to reviews

Title: Population-based prevalence survey of tuberculosis in the Tigray region of Ethiopia.

Authors:

- Gebretsadik Berhe (gebretsadik_b@yahoo.com)
- Fikre Enqueselassie (Fikreens@yahoo.com)
- Elena Hailu (elenahailu@yahoo.com)
- Wondale Mekonnen (wondalemekonnen@gmail.com)
- Tsigemariam Teklu (tsigsha@yahoo.com)
- Ataklti Gebretsadik (agsoloda@yahoo.com)
- Rezene Berhe (rezybravo@yahoo.com)
- Tewodros Haile (theohi@yahoo.com)
- Abraham Aseffa (aseffaa@gmail.com)

Version: 2 Date: 12 February 2013

Author’s response to reviews:

There was typing error of the prevalence value of females resulting in wrong calculation and the sampling scheme diagram is improved better than the previous one.